# Subacromial spacers for adults with symptomatic, irreparable rotator cuff tears: the START:REACTS novel group sequential adaptive RCT

Andrew Metcalfe,<sup>1,2\*</sup> Susanne Arnold,<sup>1</sup> Helen Parsons,<sup>1</sup> Nicholas Parsons,<sup>1</sup> Gev Bhabra,<sup>2</sup> Jaclyn Brown,<sup>1</sup> Howard Bush,<sup>2</sup> Michael Diokno,<sup>2</sup> Mark Elliott,<sup>3</sup> Josephine Fox,<sup>4</sup> Simon Gates,<sup>1,5</sup> Elke Gemperlé Mannion,<sup>1</sup> Aminul Haque,<sup>1</sup> Charles Hutchinson,<sup>1,2</sup> Rebecca Kearney,<sup>6</sup> Iftekhar Khan,<sup>1</sup> Tom Lawrence,<sup>2</sup> James Mason,<sup>1</sup> Usama Rahman,<sup>2</sup> Nigel Stallard,<sup>1</sup> Sumayyah Ul-Rahman,<sup>1</sup> Aparna Viswanath,<sup>7</sup> Sarah Wayte,<sup>2</sup> Stephen Drew<sup>2</sup> and Martin Underwood<sup>1,2</sup> on behalf of the START:REACTS team

<sup>1</sup>Warwick Medical School, University of Warwick, Coventry, UK
<sup>2</sup>University Hospitals Coventry and Warwickshire NHS Trust, Coventry, UK
<sup>3</sup>WMG, University of Warwick, Coventry, UK
<sup>4</sup>Patient Representative, Durham, UK
<sup>5</sup>Cancer Research UK Clinical Trials Unit, University of Birmingham, Birmingham, UK
<sup>6</sup>Bristol Medical School, University of Bristol, Bristol, UK
<sup>7</sup>South Tees Hospitals NHS Foundation Trust, Middlesbrough, UK

\*Corresponding author a.metcalfe@warwick.ac.uk

## **Disclosure of interests of authors**

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at <a href="https://doi.org/10.3310/TKJY2101">https://doi.org/10.3310/TKJY2101</a>.

**Primary conflicts of interest:** Andrew Metcalfe, Helen Parsons, Elke Gemperlé Mannion, Charles Hutchinson, James Mason, and Martin Underwood are co-investigators on two other NIHR-funded trials: Robotic Arthroplasty: A Clinical and cost Effectiveness Randomised controlled trial (RACER)-Knee and RACER-Hip (Andrew Metcalfe leads RACER-Knee), for which Stryker also fund treatment costs and some imaging costs. As with the presented study, the full independence of the study team is protected by legal agreements.

Andrew Metcalfe, Susanne Arnold, Helen Parsons, Nicholas Parsons, Elke Gemperlé Mannion, Aminul Haque, Charles Hutchinson, Rebecca Kearney, Iftekhar Khan, James Mason, Nigel Stallard and Martin Underwood all work on other NIHR-funded studies. Charles Hutchinson, Rebecca Kearney, James Mason and Martin Underwood are or have been members of funding panels in NIHR, although not on the EME programme. Rebecca Kearney is chair of the NIHR Programme Grants for Applied Research (PGfAR) committee, a paid position in NIHR but unrelated to the trial. She is also a previous chair of the NIHR Research for Patient Benefit (RfPB) committee and previous member of the Health Technology Assessment (HTA) Clinical Evaluation and Trials Committee and NIHR Integrated Clinical Academic (ICA) doctoral committee. Martin Underwood was a member of the NIHR Journals Library Editors Group and HTA Commissioning Committee. Until March 2021 he was an NIHR Senior Investigator. Until March 2020 he was an editor of the NIHR journal series, and a member of the NIHR Journal Editors Group, for which he received a fee. Simon Gates was a member of the NIHR Clinical Trials Unit Standing Advisory Committee and EME – Funding Committee Members and currently part of the HTA General Committee. Stephen Drew held an educational consultancy with Wright from 1 April 2016 until it was acquired by Stryker in 2021, when it migrated to an educational consultancy with Stryker from 1 April 2022.

Outside of this study, the authors report no personal financial conflict of interest with Stryker or any other related commercial organisation.

Published August 2023 DOI: 10.3310/TKJY2101

# Plain language summary

Subacromial spacers for adults with symptomatic, irreparable rotator cuff tears: the START:REACTS novel group sequential adaptive RCT Efficacy and Mechanism Evaluation 2023; Vol. 10: No. 3 DOI: 10.3310/TKJY2101

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain language summary**

Tears of the rotator cuff tendons of the shoulder are very common. Some people with a rotator cuff tear have pain and loss of function that is not improved by simple treatments, and they may undergo surgery.

Many people have the tear repaired. if this cannot be done, keyhole surgery can be used to clear space around the tendons and allow the shoulder to move better. This procedure is called an arthroscopic debridement.

The InSpace<sup>®</sup> (Stryker, Kalamazoo, MI, USA) balloon is a dissolvable device that is placed by the surgeon above the shoulder joint, after an arthroscopic debridement. We wanted to know if it improved results for patients and was good value.

We compared the standard operation, arthroscopic debridement, with the same procedure with the InSpace balloon inserted. We collected data from 117 people with rotator cuff tears from 24 NHS hospitals. Because of the COVID-19 pandemic, we could not directly measure strength or movement and used well-established questionnaires instead.

Twelve months after their operation, most people had improved. People who had the standard operation alone, without the balloon, had less pain and could use their shoulder more.

We calculated the costs of each treatment, including lost earnings. The InSpace balloon was more expensive and was poor value for money for the NHS.

We developed a new method for doing an early statistical analysis to decide whether we could stop the study early. Because of this, around half the number of people were needed in the study, compared with the number we would normally need to do this work. We did additional research and found that this new method would work for some other studies and would give much quicker results about which treatments work best.

We developed a new method for doing magnetic resonance imaging scans of the shoulder, although data collection was limited by the COVID-19 pandemic.

## **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

Efficacy and Mechanism Evaluation (EME) was launched in 2014 and is indexed by Europe PMC, DOAJ, Ulrichsweb<sup>™</sup> (ProQuest LLC, Ann Arbor, MI, USA) and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme.

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **EME** programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

#### **This report**

The research reported in this issue of the journal was funded by the EME programme as project number 16/61/18. The contractual start date was in February 2018. The final report began editorial review in April 2022 and was accepted for publication in August 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the EME programme or the Department of Health and Social Care.

Copyright © 2023 Metcalfe *et al.* This work was produced by Metcalfe *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

### NIHR Journals Library Editor-in-Chief

#### Dr Cat Chatfield Director of Health Services Research UK

### **NIHR Journals Library Editors**

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

**Dr Peter Davidson** Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Dr Rob Riemsma** Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk